Author's response to reviews

Title: Post traumatic stress symptoms and heart rate variability in Bihar flood survivors following yoga: a randomized controlled study

Authors:

Shirley Telles (shirleytelles@gmail.com)
Nilkamal Singh (naosekpamkamal@gmail.com)
Meesha Joshi (meeshie26@gmail.com)
Acharya Balkrishna (pyp@divyayoga.com)

Version: 11 Date: 24 February 2010

Author's response to reviews:

Re-orienting the tables from landscape to portrait